Dear editors,

Thanks for your hard work. I have carefully read the comments of reviewers and modified the article according to the questions raised.

Question 1. There are so much spelling mistakes in the manuscript. The article should be reconsidered from this point of view.
Answer: The grammatical structure and vocabulary of the article have been corrected, and part of the content has been rewritten.

Question 2. In the physical examination part, lung involvement can also be wondered, it would be good to provide information in this respect.
Answer: The content of lung examination has been added to the article.

Question 3. Abbreviations are not included under the table. These should be added.
Answer: Abbreviations have been added at the bottom of the table.

Question 4. References should be written in accordance with spelling rules.
Answer: References have been written in accordance with spelling rules.

Question 5. Although such cases were rare from 1989 to 2021, it is seen that the same sentences are used in the discussion part as other case presentations. In this respect, the discussion part should be re-evaluated and the authors’ own comments and sentences should be used.
Answer: The discussion section has been rewritten according to the editor’s advice.

If you have any questions, please don’t hesitate to contact me.

Yours
Chunhua Wang.